Author's response to reviews

Title: Ethics reflection groups in primary health care, an evaluation study.

Authors:

Lillian Lillemoen (lillian.lillemoen@medisin.uio.no)
Reidar Pedersen (reidar.pedersen@medisin.uio.no)

Version: 2 Date: 4 November 2014

Author's response to reviews: see over
Cover letter

Systematic ethics support in primary health care services, in terms of ethics reflection groups, has only sparsely been evaluated. The article, which is based on qualitative research data, presents the participants’ opinion and evaluation of the significance of ethics reflection groups. With reference to BMC Medical Ethics aims and scope, such as ethical aspects of clinical practice, professional choices and healthcare systems and health policies, we consider the article relevant for the journal. In our opinion the article represents a contribution to the development of knowledge about the significance of ethics reflection groups in clinical healthcare practice.